Author’s response to reviews

Title: Depressive and Anxiety Symptoms and Associated factors among Postnatal Women in Enugu-North Senatorial District, South-East Nigeria: A Cross-sectional Study

Authors:

Olaoluwa Samson Agbaje (agbiesam@yahoo.com)
Joy Anyanwu (joy.ananywu@unn.edu.ng)
Prince Iheanachor Umoke (prince.umoke@unn.edu.ng)
Tochi Iwuagwu (tochi.iwuagu@unn.edu.ng)
Cylia Iweama (cylia.iweama@unn.edu.ng)
Eyuche Ozoemena (eyuche.ozoemena@unn.edu.ng)
Ijeoma Nnaji (cdccpubhealth2010@gmail.com)

Version: 3 Date: 16 Nov 2018

Author’s response to reviews:

Abstract

The abstract has been reduced. Irrelevant details in the abstract were deleted as indicated by the editor

Background

An appropriate empirical evidence has been included in the background to indicate the prevalence of postnatal anxiety in women (See the new ref #10).

Sample size determination and procedure

The formula for sample size determination was deleted as recommended while the estimate of statistical power was computed for the study.
Data Collection

The interviewer bias inherent in the data collection process has been discussed in the limitation of the study section. Furthermore, unclear explanations in this section were deleted as recommended by the editor.

Data processing and statistical analysis

Repetitions and redundancies indicated in this section were removed. The tables are numbered in sequential order.

Strength and limitation

This section has been arranged as indicated by the editor. Information on interviewer bias and statistical power of the findings were provided

Other comment

The appendices were included after the figure legend